Author’s response to reviews

Title: The antifibrotic drug pirfenidone inhibits spondyloarthritis fibroblast-like synoviocytes and osteoblasts in vitro

Authors:

Julie Stougaard (jkl@biomed.au.dk)
Søren Lomholt (soren.lomholt@gmail.com)
Pernille Ommen (pernille.ommen.andersen@post.au.dk)
Jens Kelsen (jenskels@rm.dk)
Tue Kragstrup (kragstrup@biomed.au.dk)

Version: 6 Date: 05 Sep 2018

Author’s response to reviews:

I indeed did correct the figure. I switched the position of HLA-DR and ICAM-1 data because HLA-DR data is mentioned first. I thought this was what you meant. I am sorry that I have had such difficulties understanding your advice on this paper.

I now changed the letters in Figure 2 and corresponding sections of Results and Figure legend.